Author’s response to reviews

Title: The impact of dental treatment and age on salivary cortisol and alpha-amylase levels of patients with varying degrees of dental anxiety

Authors:
Majed AlMaummar (muammf@hotmail.com)
Huda Othman AlThabit (dr.huda.alasker@gmail.com)
Sharat Chandra Pani (sharat.pani@gmail.com)

Version: 5 Date: 13 Aug 2019

Author’s response to reviews:

1) We apologize for the confusion. The paragraph can now be found on page 15 line 40 to line 50.
Change has been made

2) Please change the Introduction heading to Background.
Change has been made

3) Please change the Methodology heading to Methods.
Change has been made

4) Please do not use bullet points in the Conclusions section, please use full sentences and paragraphs.
Bullet points have been replaced with paragraphs

5) Please include the full consent information in the Ethics approval and consent to participate section, as well as the Methods, currently it is not clear whether the consent was written or verbal in the Ethics approval and consent to participate section.
Complete information has been added to the ethics approval and consent to participate section.